Author's response to reviews

Title: Brief cognitive behavioral therapy in primary care: A patient randomized hybrid effectiveness implementation design

Authors:

Jeffrey A Cully (jcully@bcm.edu)
Maria E. A. Armento (armento@bcm.edu)
Michael R Nadorff (nadorff@bcm.edu)
Aanand D Naik (anaik@bcm.edu)
Melinda A Stanley (mstanley@bcm.edu)
Kristen Sorocco (Kristen.Sorocco@va.gov)
Mark E Kunik (mkunik@bcm.edu)
Nancy Petersen (Petersen.nancyj@med.va.gov)
Michael R Kauth (Michael.kauth@va.gov)

Version: 2 Date: 12 April 2012

Author's response to reviews:

I am uploading the additional documentation regarding our study that you have requested (IRB approval, funding, etc.).